Author’s response to reviews

Title: Migration Intentions of Asian and African Medical Students Educated in China: A Cross Sectional Study

Authors:
Wen Li (liwen_bella@xzhmu.edu.cn)
Hong Sun (sunh@xzhmu.edu.cn)

Version: 3 Date: 14 Aug 2019

Author’s response to reviews:

Dear Prof. James Buchan and reviewers,

Thank you so much for your letter and the reviewers’ comments on our manuscript entitled “Migration Intentions of Asian and African Medical Students Educated in China: A Cross Sectional Study” (HRHE-D-19-00060R2). Those comments are very helpful for revising and improving our paper, as well as the important guiding significance to other research.

 Replies to the reviewers’ comments:
Reviewer #1:
The article (particularly the background and discussion sections) need to be edited for English language. Although the authors say that the article has been proof read in English, I don't see that many language-related changes have been made since the last draft of the paper (other than those I highlighted). This needs to happen before the article is ready to be reviewed again.

Response:
We have now thoroughly gone through the manuscript and made substantial corrections this time (the changes are highlighted in green).

Thank you again for your constructive comments and suggestions which would help us both in English and in depth to improve the quality of the paper.

Kind regards,
Wen Li
E-mail: liwen_bella@xzhmu.edu.cn

Corresponding author: Prof Hong Sun
E-mail: sunh@xzhmu.edu.cn